Author’s response to reviews

Title: Sedentary Behaviour in hospitalised older people: a scoping review protocol

Authors:

Unyime Jasper (unyime.jasper@adelaide.edu.au)
Lalit Yadav (lalit.yadav@adelaide.edu.au)
Agathe Daria Jadczak (agathedaria.jadczak@adelaide.edu.au)
Solomon Yu (solomon.yu@adelaide.edu.au)
Renuka Visvanathan (renika.visvanathan@adelaide.edu.au)
Joanne Dollard (joanne.dollard@adelaide.edu.au)

Version: 2 Date: 15 Jan 2020

Author’s response to reviews:

Reviewer 1:
Thank you for your insightful comments and recommendations. We have amended the protocol according to your comments. We have taken a “step back” to identify the key domains of sedentary behaviour and sedentary behaviour bouts. Changes are reflected all through the protocol. The purpose of the scoping review now reads “Thus, the primary aim of this review is to describe the state of the peer-reviewed literature on key domains of SB and SBB in hospitalised older people”.

Reviewer 2:
Many thanks for your insightful comments and suggestions.
The aim now reads “Thus, the primary aim of this review is to describe the state of the peer-reviewed literature on key domains of SB and SBB in hospitalised older people”.
The limit (S15 and S16) have been removed from the new search as advised.